Edmonton School District No 7 (2007 - 2012)

THIS AGREEMENT is between the Board of Trustees of Edmonton School District No 7 and the Alberta Teachers' Association, a body corporate incorporated under the laws of the Province of Alberta.

Whereas the Association is the duly certified bargaining agent for the teachers employed by the Board, and

Whereas such teachers' terms and conditions of employment and their compensation have been the subject of negotiation between the parties;

Therefore, the parties desire that these matters be set forth in an agreement to govern all the terms of employment of the teachers.

1. Definitions

For the purposes of this agreement:

(a) "Benefits" shall mean the Board's portion of group insurance premiums.

(b) "Board Cost" for a teacher shall mean applicable salary and benefits as well as the Board's obligation for holiday pay, vacation pay and the Board portion of legislated programs such as EI and the Canada Pension Plan.

(c) "Local" shall mean the Edmonton Public Teachers Local 37 of the Alberta Teachers' Association.

(d) "Natural Break" means Christmas break, spring break, summer break, other breaks of four or more week days, semester changes at high schools and any other time approved by the Board.

(e) "Salary" shall mean, unless otherwise specified, applicable salaries and allowances specified under clauses 5.1 and 7 to 11 inclusive.

(f) "School Year" shall mean the period beginning on the first day of operation and ending on the day prior to the first day of operation in the following year.

(g) "Substitute Teachers" are teachers employed on a day-to-day basis and placed on a list of active substitute teachers.

(h) "The Association" shall mean the Alberta Teachers' Association, a body corporate incorporated under the laws of the province of Alberta.

2. Term and Duration

2.1 This agreement shall, except where otherwise specified, take effect on September 1, 2007 and shall remain in full force and effect until August 31, 2012, except in the event that negotiations between the parties for renewal or revision hereof or for a new collective agreement have been commenced as hereinafter provided and have not been concluded by August 31, 2012, this agreement shall continue in full force and effect until:

(a) a new agreement is reached, or

(b) the Association terminates the agreement by commencement of strike action, or

(c) the Board terminates the agreement by the commencement of a lockout.

2.2 No new claims or demands shall arise out of previous agreements upon the coming into force of this agreement.

2.3 In the event that any provision of this agreement is unenforceable or invalid, that unenforceability or invalidity shall not affect the enforceability or validity of any other provisions of this agreement.

3. Notice to Bargain

3.1 Either party to the collective agreement may, not less than 60 days and not more than 150 days preceding the expiry of the term of the collective agreement, by notice in writing, require the other party to the collective agreement to commence collective bargaining. When the foregoing notice is given, the representatives of the parties shall meet and commence collective bargaining not more than 30 days after notice is given.

3.2 The two parties may at any time upon their mutual agreement negotiate revisions of this agreement. Any such revisions shall become effective from such date as shall be mutually agreed upon by the parties.

4. Scope

4.1 This agreement shall be applicable to every person who requires a teacher certificate as a condition of employment with the Board, specifically exempting the superintendent of schools, assistant superintendents, managing directors, directors and supervisors of personnel.

4.2 Notwithstanding the above, this agreement shall not apply to persons teaching at the Board's summer school, night classes or tutorial classes except as provided in clause 11.2. These staff shall only have access to clause 25 (Grievance Procedure) for issues arising out of clause 11.2.

4.3 The Board may create new classes of designation in respect to teachers covered by this agreement. Nevertheless, the salaries and allowances for such new classifications shall be arrived at by consultation with the Local before initially advertising positions or designating teachers within the new classifications.

5. Salary Scale

5.1 Effective September 1, 2007 teachers will be paid in accordance with the following salary grid.

Years of teaching experience Years of University Education

	Four	Five	Six
0	48,789	51,583	54,743
1	51,571	54,365	57,526
2	54,353	57,149	60,308
3	57,135	59,931	63,090
4	59,917	62,713	65,872
5	62,700	65,495	68,654
6-7	65,482	68,277	71,436
8	68,264	71,059	74,218

9	71,046	73,841	77,000
10	73,828	76,623	79,783
11	77,196	79,992	83,151

Effective September 1, 2008 step 6/7 becomes 7/8 Effective September 1, 2009 step 7/8 becomes 8/9 Effective September 1, 2010 step 8/9 becomes 9/10 Effective September 1, 2011 step 9/10 becomes 10/11 Effective September 1, 2008, 2009, 2010 and 2011:

Salaries shall be increased on September 1 of the years 2008 through 2011, by the average weekly earnings index in Alberta from the previous calendar year, as specified in Appendix B of the Memorandum of Agreement between the Alberta Teachers' Association and the Government of Alberta dated November 15th, 2007.

Accordingly the increases shall be as follows:

September 1, 2008: Average Weekly Earning Index increase in Alberta from previous calendar year (Jan-Dec 2007 over 2006)

September 1, 2009: Average Weekly Earning Index increase in Alberta from previous calendar year (Jan-Dec 2008 over 2007)

September 1, 2010: Average Weekly Earning Index increase in Alberta from previous calendar year (Jan-Dec 2009 over 2008)

September 1, 2011: Average Weekly Earning Index increase in Alberta from previous calendar year (Jan-Dec 2010 over 2009)

5.1.1 Teachers on staff as of September 1, 1991 who have one or two years of teacher education shall be paid the same grid salary as teachers with four years of teacher education and two years of teaching experience until they attain four years of teacher education. This clause shall be deleted by mutual consent only after the District and the Association are satisfied that there are no teachers eligible for payment under this clause.

5.1.2 Teachers on staff as of September 1, 1991 who have three years of teacher education shall be paid the same grid salary as teachers with four years of teacher education and four years of teaching experience until they attain four years of teacher education. This clause shall be deleted by mutual consent only after the District and the Association are satisfied that there are no teachers eligible for payment under this clause.

5.1.3 New teachers with less than four years university education shall be paid at the category four, step zero level until they qualify for category four on the grid as evaluated by the Teacher Qualifications Service (TQS) in accordance with clause 5.5.

5.2 Salary Payment

5.2.1 The Board shall pay by electronic deposit to the financial institution of each teacher's choice taking into consideration any necessary adjustments:

(a) 1/12 of annual salary by the last banking day of each month except for summer break and Christmas/spring breaks.

(b) June and July salary will be paid on the last banking day of June. Teacher payout for August shall be paid on or before the fifth working day in July.

(c) Salary payments prior to the Christmas and spring breaks shall be deposited by the last operational day prior to the break.

5.2.2 A teacher may request in writing that the Board, through payroll deductions, make electronic RRSP deposits to the financial institution of the teacher's choice, in the amount specified by the teacher.

5.2.3 Administrators', supervisors' and specialists' allowances shall be paid from the date the respective responsibility is assumed and shall be discontinued from the date the responsibility is terminated. However, clauses 8(b) and 8(c) apply.

5.2.4 The Board shall ensure that the pensionable service of teachers is not negatively impacted by transferring to and from schools with modified calendars. Upon request, a teacher newly-appointed to the District and assigned to a school with a modified calendar shall receive an advance repayable over the term of the teacher's contract or the school year, whichever is shorter, or other mutually agreed upon arrangements between the teacher and the Board.

5.3 New Appointees

5.3.1 A teacher may be placed at one or more steps on the salary grid higher than that provided for under teaching experience when the teacher is engaged by the Board for the purpose of filling:

(a) a specialized teaching position and/or supervisory position.

(b) a teaching and/or supervisory position which, in addition to an Alberta Teacher Certificate, requires a special license or certificate, the training for which is not recognized by the Alberta Teachers' Association Teacher Qualifications Service.

The grid salary so determined shall not exceed the maximum provided for the appropriate year of teacher education.

5.3.2 Prior to the engagement of a teacher in accordance with clause 5.3.1, the Board shall notify the Association of its intention, give the Association one week to respond and promptly notify the executive assistant of the Local when the appointment has been made.

5.4 Experience Increments

5.4.1 Teaching experience for the purpose of clauses in this section shall mean experience gained while holding a valid Alberta teaching certificate or its equivalent, and includes the following:

(a) days under contract in a position which requires a teaching certificate as a condition of employment, excluding leaves of absence without salary and vacation periods;

- (b) days employed as long-term substitute teacher; (on grid as per clause 7.2)
- (c) days employed as a day-to-day substitute teacher within the preceding five years;
- (d) days taught for the Alberta Distance Learning Centre;
- (e) days taught in a post-secondary institution.

5.4.2 Each teacher who is eligible for an increment shall, on the first day of September or February immediately following the establishment of such eligibility, be placed on the next higher step on the grid for the appropriate category of teacher education.

5.4.3 Initial placement on the salary grid for teachers with temporary or probationary contracts and substitute teachers, shall be determined using the following criteria.

(a) any increments previously granted by this Board for the teacher will be recognized.

(b) one increment will be given for each full year of teaching experience, as experience is defined in clause 5.4.1.

(c) where service includes for following:

- substitute or supply service;

- partial years of service; and

- periods of part-time service,

full years of service shall be calculated as the number of full day equivalents of teaching experience in those years divided by 130.

(d) any residual number of days of teaching experience beyond the full years of teaching experience recognized above shall be credited to the teacher for accumulation toward the earning of any future increment.

5.4.4 For teachers with a continuous contract, experience increments shall be determined using the following criteria:

(a) teachers who have 130 full days of teaching experience which have not previously been counted for increment purposes shall be granted one experience increment.

(b) thereafter, an additional increment will be granted for each year of service with a minimum of 130 full days of teaching experience acquired until the teacher reaches the maximum grid salary of the appropriate category of education.

(c) these increments will be effective on September 1 or February 1 immediately following the attainment of the necessary accumulation of days.

(d) a teacher shall be granted only one experience increment during any one school year. However, a teacher who earns an increment while on Professional Improvement Leave (clause 23), shall retain the right to earn a further increment in February following the return from leave.

(e) Teachers with a part-time continuous contract will earn one increment for every 130 days of service. The increment will be granted on the subsequent September 1 or February 1, whichever occurs first. Residual days shall carry forward.

(f) the maximum number of days of teaching experience which may be counted for increment purposes during any school year shall be 130 days.

5.4.5 Teachers moving from a 12 month position back to a 10 month position shall be granted one year of experience for each full year of service. Those teachers who work part time on a 12 month contract shall have their experience prorated as follows:

#FTE days = X worked

230 130

X represents number of days of experience credited to the teacher.

5.5 Teacher Education

5.5.1 The Alberta Teachers' Association Teacher Qualifications Service shall evaluate a teacher's education for grid salary purposes in accordance with the policies and principles approved by the Teacher Salary Qualifications Board established by memorandum of agreement among Alberta Education, Alberta Teachers' Association and the Alberta School Trustees' Association dated March 23, 1967.

5.5.2 The effective dates of adjustment to grid salaries for increased teacher education shall be the opening day of school in each school year and February 1 each year.

5.5.3 A teacher must advise the Board in writing if a further year of teacher education is acquired as verified by the Teacher Qualification Service (TQS). Failure to do so by October 31 will result in the grid salary adjustment date to be the following February 1. Failure to advise the Board in writing by March 31 will result in the grid salary adjustment being delayed until the opening day of school in the next school year.

6. Work During Holiday Periods

6.1 With the exception of school administrators, a teacher covered by this collective agreement who agrees to render services during the summer, Christmas, spring recess periods or other natural breaks at the request of the superintendent of schools, shall be paid 1/400 of total annual salary for each half day of work, or may choose an equal number of days in lieu of payment. Such days will be taken at a time suitable to the teacher, subject to approval of the superintendent of schools, having regard to all the circumstances and the interest of the district.

6.2 Days in lieu earned under the provisions of clause 6.1 must be taken by June 30 of the school year following the one in which they were earned. Days in lieu not taken by this date shall be paid out by September 30 of the subsequent school year at a rate of 1/400 of the teacher's total annual salary for each 1/2 day of work.

6.3 Clause 6.1 shall apply to school administrators, in circumstances deemed by the Superintendent to be beyond the proper operational parameters of their schools.

6.4 Teachers who, with the approval of the superintendent of schools, undertake and complete discrete projects will be paid the sum of money assigned to the project for teacher services.

7. Substitute Teachers

7.1 The rate of pay for substitute teachers shall be, inclusive of general holiday and vacation pay, as follows:

	Sept 1/07
Full Day	\$180
Half	\$99

Day

The half day rate of pay for substitute teachers shall be adjusted as follows:

September 1, 200856% of the full day rateSeptember 1, 200958% of the full day rateSeptember 1, 201060% of the full day rate

Increases to this allowance will be consistent with the date and amount of increases to the salary grid in clause 5.1.

7.2 Notwithstanding clause 7.1, a substitute teacher who substitutes for the same teacher for a period of more than one consecutive teaching day shall be paid 1/200 of the appropriate grid placement for the substitute teacher from the beginning and during the continuance of such consecutive teaching days.

7.3 The Board may, on application by a substitute teacher, reimburse that teacher for medical expenses resulting from injury on the job.

7.4 If a substitute teacher is unable to work as a result of an injury incurred at the workplace, the Board shall pay the teacher the per diem rate specified in clause 7.1 for a maximum of 20 consecutive teaching days immediately following the injury, provided that the inability to work is verified by a physician chosen or approved by the Board. It is understood that any related medical expenses incurred by a substitute teacher in accessing the verification by a physician of the Board's choosing will be paid for by the Board.

8. Allowances and Salaries for Designated Positions

(a) A teacher is eligible to receive only one allowance with the exception of work experience coordinator and/or registered apprenticeship coordinator or a teacher assigned to multiple locations. It is understood that in the application of this clause, no teacher shall be assigned the duties of, or be paid the accompanying allowances for more than two designations.

(b) A teacher in receipt of an allowance under clauses 10, 9.4 and 9.3 shall, following three consecutive years in the designation and upon acceptance of another position in the agreement, retain the same salary for a period of two years, or until the salary of the new position exceeds the amount of the retained salary, whichever occurs first.

(c) The salary of a principal or assistant principal shall:

(i) upon termination of designation and acceptance of another position in the agreement, remain the same for a period of two years or until the salary of the new position exceeds the amount of the retained salary, whichever occurs first, and

(ii) not be reduced because of declining enrolments to an amount less than the salary to which the individual was entitled by the formula or schedule in the preceding school year.

9. Allowances for Designated Positions

In addition to the grid salary, there shall be paid the following annual allowances to designated personnel in the school system. All such allowances shall be prorated for teachers employed less than full-time with the Board.

9.1 Principals

The formula to be used in computing the principal's allowance for each school will be as follows:

Sept 1/07 Minimum \$20,600 + \$14.68 (P-300)

Maximum \$32,159

Increases to this allowance will be consistent with the date and amount of increases to the salary grid in clause 5.1.

For the purpose of this clause, "P" shall be defined as the sum of the per pupil allocation, based on the September 30 enrolment count for the school, divided by the basic allocation for a full-time regular elementary student.

Upon request by the Executive Assistant of the Local, the Board shall provide the following for each school: a) the value of P and b) the September 30 student enrolment count.

Principals in non-school administrative assignments shall be paid an allowance calculated on the basis of a 500 full-time regular elementary student count. However, clause 8(c) applies.

9.2 Assistant Principals

Teachers who are designated as assistant principals shall be paid 50 per cent of the principal's allowance.

Teachers who are designated as assistant principals in non-school administrative assignments will be paid 50 per cent of an allowance calculated on the basis of a 500 full-time regular elementary student count. However, cause 8(c) applies.

9.3 Curriculum Coordinators and Department Heads

Sept

1/07

\$6,190

Increases to this allowance will be consistent with the date and amount of increases to the salary grid in clause 5.1.

Assistant Curriculum Coordinator and Assistant Department Heads

Sept

1/07

\$4,323

Increases to this allowance will be consistent with the date and amount of increases to the salary grid in clause 5.1.

9.4 Teacher Consultants

Sept 1/07

\$6,190

Increases to this allowance will be consistent with the date and amount of increases to the salary grid in clause 5.1.

9.5 A teacher not in receipt of an allowance listed in the preceding clauses shall receive 1/400 of the principal's allowance for each half day they are appointed as the principal designate.

10. Salaries for Designated Positions (12 months)

Teachers in the following positions shall work a 12-month year with six weeks vacation and receive the following annual salaries.

10.1 Directors receive a salary equal to the maximum principal allowance provided for in clause 9.1 plus the maximum salary for six years provided for in the salary grid in clause 5.1.

Sept 1/07

\$115,310

Increases to this allowance will be consistent with the date and amount of increases to the salary grid in clause 5.1.

10.2 Supervisors receive a salary equal to 90 per cent of a director's salary.

Sept 1/07

\$103,779

Increases to this allowance will be consistent with the date and amount of increases to the salary grid in clause 5.1.

10.3 Twelve Month Consultants receive a salary equal to 90 per cent of a supervisor's salary.

Sept 1/07

\$93,401

Increases to this allowance will be consistent with the date and amount of increases to the salary grid in clause 5.1.

10.4 Program Coordinators receive a salary equal to a maximum assistant principal's allowance plus the maximum salary provided for on salary grid in clause 5.1.

Sept 1/07

\$99,231

Increases to this allowance will be consistent with the date and amount of increases to the salary grid in clause 5.1.

11. Other Allowances

11.1 Teachers Assigned to Multiple Locations

A teacher whose assignment requires travel between buildings separated by at least one kilometre shall be paid an annual allowance as follows:

Sept 1/07

\$1,189

If a teacher is regularly scheduled to teach in more than 2 buildings in a single day, there shall be an additional allowance as follows:

Sept 1/07

\$1,189

Increases to this allowance will be consistent with the date and amount of increases to the salary grid in clause 5.1.

11.2 Teachers at Night School and Summer School

A teacher employed on an hourly basis to provide instruction in credit courses at the Board's night school and summer school shall be paid at a rate inclusive of general holiday and vacation pay of:

Sept

1/07

\$54.51

Increases to this allowance will be consistent with the date and amount of increases to the salary grid in clause 5.1.

11.3 Work Experience and/or Registered Apprenticeship Program Coordinator

Sept 1/07

\$2,601

Increases to this allowance will be consistent with the date and amount of increases to the salary grid in clause 5.1.

12. Group Insurance

12.1 Participation in group insurance plans shall be a condition of employment of all new appointees.

12.2 Notwithstanding clause 12.1, the requirement to participate in either Alberta Health Care, extended health care, dental care or vision plan insurance, as a condition of employment, shall be waived for those teachers who already have such group insurance coverage as dependents of their spouses and who therefore elect not to participate. Further, teachers with no dependents other than spouses may elect to take single coverage in any of Alberta Health Care, extended health care, dental care or vision plan insurance plans if their spouses have single coverage in the same or comparable plans.

12.3 The Board's contribution to group insurance premiums for teachers employed by the Board shall be at the rate stipulated in the following table:

Alberta Health Care Insurance 100%

Alberta School Employee Benefit Plan

100%

Extended Disability (Plan D) - The Board contribution is 80%

Effective September 1, 2010 the Board contribution to the Extended

Disability Premium (Plan D) will be changed to

85%

Dental Care (Plan 3C) and Vision (Plan 3)

Effective September 1, 2007 the Board contribution to the Dental plan and Vision Plan is a cash amount equal to the premiums in effect for the 2007-08 school year.

Effective September 1, 2008 the Board contribution to the Dental plan and Vision Plan is a cash amount equal to the premiums in effect for the 2008-09 school year.

Effective September 1, 2009 the Board contribution to the Dental plan and Vision Plan is a cash amount equal to the premiums in effect for the 2009-10 school year.

Effective September 1, 2010 the Board contribution to the Dental plan and Vision Plan is a cash amount equal to the premiums in effect for the 2010-11 school year.

Effective September 1, 2011 the Board contribution to the Dental plan and Vision Plan is a cash amount equal to the premiums in effect for the 2011-12 school year.

Extended Health Care (Plan 1) 100%

12.3.1 Health Spending Account -

Effective September 1, 2007, the board will contribute annually an amount of \$550 for each 1.0 FTE teacher. Effective September 1, 2008, the board will contribute annually an amount of \$600 for each 1.0 FTE teacher. Effective September 1, 2009, the board will contribute annually an amount of \$650 for each 1.0 FTE teacher. Effective September 1, 2010, the board will contribute annually an amount of \$700 for each 1.0 FTE teacher. Effective September 1, 2011, the board will contribute annually an amount of \$750 for each 1.0 FTE teacher.

This contribution shall be prorated for teachers employed less than full time with the Board. The unused balance will be carried forward for a total accumulation of two years. Teachers leaving the employ of the Board will forfeit any remaining balance.

12.4 A teacher employed under a contract which terminates on the date in June that school closes shall have all benefits or pay in lieu of benefits received through this collective agreement extended until the end of the school year as defined.

12.5 A teacher who suffers personal injury arising out of and in the course of employment and who incurs medical expenses not covered in group insurance plans sponsored by the Board or covered by other government agencies shall be entitled to reimbursement for such reasonable expenses upon presentation of receipt for bills paid. Notwithstanding the above, the Board shall not be liable for payment of costs beyond the period of one year from the date of the accident that caused the injury.

12.6 Notwithstanding the above, for teachers who are in receipt of an ATRF pension who are employed under a contract and are not eligible to enroll in ASEBP, the Board shall pay to each teacher the equivalent amount they would

have contributed had the teacher participated in all plans. The Board shall pay single or family coverage based upon the teacher's Alberta Health Care enrollment.

13. Teacher Assignment

13.1 Teachers have the right to assist in determining the grouping of students for instructional purposes and in determining instructional duties and other duties in accordance with Board policies and the terms of this agreement. It is the responsibility of each teacher to provide such instruction and perform such duties as assigned by the principal.

13.2 A teacher will not be assigned duties in excess of 1,800 minutes per week, of which a maximum of 1430 minutes will be devoted to the instruction of pupils. A teacher may agree to be timetabled for instructional duties which may vary in the number of minutes assigned per week. A teacher shall not be assigned instructional duties which would exceed an average of 1430 minutes per week for any given school year.

13.2.1 Assignable time will provide for instruction, supervision of students and professional activities such as inservice sessions, staff meetings, committee work and parent/teacher conferences.

Parent/teacher conferences, unless otherwise agreed to by the teacher, will be scheduled on operational days, excluding Fridays, in a way that will:

(a) recognize the teacher's total workload, including assigned duties and other professional responsibilities,

(b) accommodate the reasonable requests of parents, and

(c) provide a reasonable work day for teachers.

The parties agree that, effective September 1, 2012, the language of clause 13.2 will revert to the language of the September 1, 2001 collective agreement.

The language reverted to is set out below:

13.2 A teacher will not be assigned duties in excess of 1,800 minutes per week. Assignable time will provide for instruction, supervision of students and professional activities such as inservice sessions, staff meetings, committee work and parent/teacher conferences.

Parent/teacher conferences, unless otherwise agreed to by the teacher, will be scheduled on operational days, excluding Fridays, in a way that will:

(a) recognize the teacher's total workload, including assigned duties and other professional responsibilities,

(b) accommodate the reasonable requests of parents, and

(c) provide a reasonable work day for teachers.

The implementation of the freeze on the sunsetting of the "hours of work/minutes of instruction" which is referenced in the November 15, 2007 Memorandum between the Province of Alberta and the Alberta Teachers' Association will operate on a without prejudice basis and will not be considered to be a precedent to adversely affect the rights of either the Edmonton Public School District or the Alberta Teachers' Association for any purpose.

13.3 The instruction of a single junior high school class in a given course by two or more teachers will occur only with the concurrence of the teachers involved.

13.4 A teacher who believes that the instructional groupings or assigned duties are unreasonable may request a review by the superintendent of schools. In such a case, the superintendent of schools shall promptly cause an investigation to be made and on the basis of such investigation, shall promptly determine whether or not the assignment is reasonable. The reasons for the determination shall be promptly provided in writing to the teacher.

13.5 A teacher who is transferred from one school to another at the initiative of the Board shall be given reasons in writing.

14. Time for Administration and Supervision

14.1 Principals, assistant principals and other administrative personnel in each school shall be allowed such time for administration and supervision as the superintendent of schools may determine but not less than the following schedules:

14.2 Elementary Schools

No of Classrooms as of September 30	No of Days Per Week
3 - 5	1/2 day
6 - 7	1 day
8 - 9	1 1/2 days
10 - 13	2 days
14 - 16	3 days
17 - 19	4 days
20 – 30	5 1/2 days
31 and over	6 days
14.3 Elementary/Junior Hig	h Schools
No of Classrooms as of September 30	No of Days Per Week
4 - 5	1 day
6 - 7	1 1/2 days
8 - 9	2 days
10 - 13	2 1/2 days

14 - 16	3 1/2 days
17 - 19	4 1/2 days
20 - 25	5 1/2 days
26 - 32	6 days
33 and over	6 1/2 days

14.4 Senior High Schools

The principals shall be granted full time for supervision and administration but may elect to teach for a portion of their administrative time. In addition, there shall be granted to each school, supervision and administration time for assistant principals in accordance with the following table:

Size of School as of September	Number of Periods Based on
30	an Eight Period Day
Over 1,500 registrations	50
1,300 - 1,499 registrations	45
1,100 - 1,299 registrations	40
900 - 1,099 registrations	35
700 - 899 registrations	30

14.5 Provided, always, that the apportionment of the aggregate administrative and supervisory time allotted may be varied at the discretion of the principal with the concurrence of the assistant(s) and the approval of the superintendent of schools.

14.6 For the purpose of determining number of classrooms in a school, career and technology studies (CTS) labs shall be counted as classrooms.

15. Administrative Absences

15.1 In the event that any incumbent of an administrative position in a school is absent from duty for a period in excess of five consecutive teaching days, another administrator, supervisor or teacher shall assume the responsibility and be paid the allowance of the administrative position the teacher temporarily occupies, commencing with the sixth day. When an assistant principal assumes the responsibility of the principal in the same school in accordance with the above

procedure, another teacher shall immediately assume the responsibilities and be paid the allowance of the assistant principal.

15.2 In a school where there is no assistant principal, a teacher shall be named to carry out administrative duties during any absence of the principal.

The principal may, for the school year, designate up to two teachers who are not in receipt of any other allowances to serve for the school year as surrogate principal during the principal's absence. There shall be a surrogate principal's annual allowance of \$1,236 effective September 1, 2007.

Increases to this allowance will be consistent with the date and amount of increases to the salary grid in clause 5.1.

15.3 Payments under this clause will be made annually by the last banking day of June.

16. Staffing

16.1 Both parties to the collective agreement recognize the Board's responsibility to determine the staffing pattern for the system.

16.2 It is also agreed that, to provide an effective educational program, a degree of flexibility to the assignment of the basic classroom staff is required. The amount of flexibility will reflect the financial resources available and the needs of individual schools.

16.3 In the event that a reduction in teaching staff is necessary as a result of declining enrolment, the Board favors the principle of natural attrition to effect this reduction.

17. Personnel Files

17.1 Teachers shall, upon request, be given access to the contents of their personnel file.

17.2 The Board shall provide to the Association a list of teachers covered by this agreement identifying name, location, FTE, position and teaching certificate number. This shall be provided in electronic format on November 15 and April 15 of each year.

18. Leaves of Absence

18.1 General Leaves of Absence

18.1.1 Reasonable requests for leave and for extensions to a leave will be granted by the superintendent of schools, having regard to all of the circumstances and the interests of the school and district. Such leaves and extensions may be for any purpose, including those purposes dealt with elsewhere in this agreement.

18.1.2 Leave granted under clause 18.1.1 will be:

(a) with salary and benefits, or

(b) with salary and benefits less the rate of pay of a substitute teacher whether or not a substitute teacher is required, or

- (c) without salary but with benefits, or
- (d) without salary or benefits.

18.1.3 Upon return to regular duty, the teacher may be required by the superintendent of schools to furnish evidence of compliance with the terms of the agreement under which the leave was granted. Failure to provide this information shall entitle the Board to be reimbursed for the salary or allowance paid.

18.1.4 (a) A teacher granted a leave of absence for a school year under clause 18.1.1 will, by March 15 of that year, notify the superintendent of schools of the teacher's intentions for the following school year.

(b) A teacher who does not respond by that date will be sent a letter by registered mail to an address agreed upon by the teacher and the Board at the commencement of the leave indicating that the teacher must, within 60 days of the date the letter is mailed, advise the superintendent of schools whether or not the teacher will be returning to duty at the beginning of the following school year. Copies of the registered letters will be sent forthwith to the executive assistant of the Local.

(c) If a teacher does not respond within the 60 day limit, that teacher's contract of employment will be deemed to be terminated by mutual consent.

18.2 Sick Leave

18.2.1 Sick leave with salary and benefits will be granted to a teacher for the purpose of obtaining necessary medical or dental treatment or because of medical disability.

18.2.2 A teacher on interim or probationary contract shall be provided 20 working days of sick leave entitlement.

A teacher employed on a temporary contract shall have such sick leave entitlement equivalent to the number of days taught in the school year divided by nine.

18.2.3 A teacher on continuing contract shall be provided 90 calendar days of sick leave entitlement.

18.2.4 After 90 calendar days of continuous absence due to medical disability, no further salary or benefits will be paid except for the following circumstances:

(a) A teacher in receipt of payment under extended disability insurance shall be entitled to the Board portion of insurance premiums.

(b) A teacher who no longer qualifies for extended disability insurance and is unable to resume his or her duties due to medical disability shall be entitled to the Board portion of insurance premiums for a period of 90 calendar days following the termination of extended disability insurance payments.

18.2.5 A teacher who has been absent for the above reasons and returns to regular duties shall have the 90 calendar day sick leave entitlement reinstated. However, in instances where the teacher has been continuously absent for a period of 60 or more calendar days, reinstatement of the sick leave entitlement shall be made contingent on the teacher providing a medical certificate, signed by a medical practitioner, approved or selected by the Board, prior to the date of return, verifying that the teacher is able to return to work on a continuing basis.

18.2.6 In order to qualify for payment of sick leave:

(a) when the sick leave is for a period of three days or less, a teacher shall provide, on a form to be supplied by the Board, a declaration as to the reason for the absence under clause 18.2, or

(b) when the sick leave is for a period in excess of three consecutive teaching days, a teacher shall provide, upon the request of the superintendent of schools, a certificate signed by a medical practitioner indicating that the absence was necessitated by medical disability.

18.2.7 The Board shall be entitled to require of a teacher but at no cost to the teacher, a medical examination by a medical practitioner or dentist selected by the Board. Such an examination, where practical, will occur on an operational

day. The Board shall ensure that the medical practitioner or dentist is requested to provide a copy of any resultant report to the teacher.

18.2.8 The teachers covered by this collective agreement waive any claims to rebates under the provisions of the Employment Insurance Act.

18.2.9 Teachers unable to carry on their duties on account of illness or any other cause shall give at least one hour notice to personnel services before school assembles so that substitute teachers may be obtained. Before returning to duty, the absentee teacher shall also notify personnel services of such intended return. If returning for the morning session, the notification must be given at least one hour before school opening, and for the afternoon session, two hours before classes assemble. If a teacher fails to observe this requirement, the superintendent of schools may, after investigating the circumstances, charge the teacher 1/2 day's rate of pay of a substitute teacher (as per clause 7.1) if one was provided.

18.3 Family Illness and Bereavement Leave

18.3.1 Leave necessitated by the critical illness or the death of a spouse, child, parent, brother, brother-in-law, sister, sister-in-law, parent of spouse, son-in-law, daughter-in-law, grandparents, grandchildren or a member of the teacher's household, shall be granted by the superintendent of schools with salary and benefits:

- (a) up to and including five operational days for critical illness;
- (b) up to and including five operational days for death; or
- (c) up to and including 10 operational days for combined critical illness and death.

18.3.2 Leave granted under clause 18.3.1(a) will terminate in the event of death of the person identified in clause 18.3.1 and the teacher will then be eligible for leave under clause 18.3.1(b). Leaves referred to in clauses 18.3.1(a) and (b) shall be taken within a two week period of the actual occurrence of the critical illness or death. Leave under 18.3.1(c) shall be taken within a three week period of the actual occurrence of the critical illness and death.

18.3.3 Before payment is made for leave under clause 18.3.1 (a), the superintendent of schools may require a medical certificate stating that critical illness was the reason for the absence.

18.3.4 Leave up to one day necessitated by the death of an individual other than those referred to in clause 18.3.1 shall be granted with salary and benefits.

18.3.5 Leave with pay and benefits up to three days per school year necessitated to attend to the medical needs of a parent, spouse's parent, child or a member of the teacher's household shall be granted by the superintendent of schools.

18.4 Private Business Leave

18.4.1 Leaves of absence for private business may be granted by the superintendent of schools, having due regard to all the circumstances and the interests of a school and/or the school system, for up to two days per school year.

18.4.2 Leave granted shall be with salary and benefits less the rate of pay of a substitute teacher as provided for in clause 7.1.

18.4.3 Requests for leave which would extend the Christmas, spring recess, summer vacation period and other breaks of four or more week days may not be granted under this clause.

18.4.4 A teacher shall be permitted to accumulate and use any unused days up to a maximum of three days.

19. Other Leaves of Absence

19.1 A teacher who is appointed to the interpretations committee under clause 25.1 step four shall be granted leave with salary and benefits for the purpose of attending such committee meetings.

19.2 A teacher who is elected to the ATA Local teachers' negotiating subcommittee shall be granted leave with salary and benefits, less the cost of a substitute teacher, for the purpose of meeting with the Board representatives to negotiate a collective agreement. The rate of pay of the substitute teacher shall be paid by the Local.

19.3 A teacher elected to represent the Local at the annual meeting of the Association shall be granted leave of absence with salary and benefits for the purpose of attending such a meeting. The rate of pay of a substitute teacher, whether required or not, will be paid by the Association.

19.4 Teachers who are elected or who are appointed by the executive to serve on committees of Local 37 ATA may be granted leave of absence with salary and benefits at the request of the Local. If such leave is granted, the Local will pay the rate of pay of a substitute teacher, as provided for in clause 7.1, whether or not a substitute teacher is required.

19.5 At the request of the Local, a teacher who is elected to the office of president of Local 37 ATA shall be granted leave of absence on a scheduled basis, up to a maximum of full-time, for the school year(s) during which the office is held. During such scheduled leave of absence, the president shall receive all the entitlements accorded to a continuing full-time teacher and the Local shall pay a prorated portion of Board cost which reflects the amount of release time provided.

19.6 In the event a teacher resigns designation to accept an elected position in the Association, the teacher shall, at the beginning of the school year following service in that capacity, be redesignated to a position no less favorable than the one that the teacher left. This provision shall apply for a maximum two year period but may be extended at Board discretion.

19.7 Leave of absence without loss of salary and benefits shall be granted

(a) for jury duty or any summons related thereto, or

(b) to answer a subpoena or summons to attend any court proceeding as a witness in a cause other than the teacher's own. The teacher shall reimburse the Board an equivalent amount of any witness fee or jury stipend set by the court.

19.8 A teacher who is granted leave in accordance with a deferred salary leave plan approved by the Board, shall upon return to duties, be entitled to return to the school to which the teacher was assigned when the leave commenced, on the same basis as continuing members of that staff.

Notwithstanding the above, a teacher, in receipt of an allowance under clauses 8, 9 and 10 for designations which continue from year to year, who is granted a leave in accordance with a deferred salary leave plan approved by the Board, shall be granted such leave from designation but not from location.

20. Maternity Leaves and Entitlements

20.1 Maternity Leave

20.1.1 Teachers with temporary, probationary or interim contracts may access personal leave for maternity reasons under clause 18.1.2(d) for up to one year from the date of delivery or until the completion of their contract, whichever occurs first. This leave may begin prior to, but not later than, the expiration of the teacher's sick leave entitlements as per 20.2.1.

20.1.2 Maternity leave in accordance with clauses 20.1.3 to 20.1.8 shall be granted to a teacher who is currently on contract and who has provided one year of consecutive service under an interim, temporary, probationary and/or continuing contract in the year preceding the starting date of the leave. "Natural Breaks" as defined in clause 1(d) and periods of time up to a total of six weeks when the teacher is not under contract, shall not be considered an interruption of consecutive service in the application of clause 20.1.2.

20.1.3 Maternity leave shall be for a minimum of 15 weeks. Maternity leave combined with parental leave shall be for a maximum of 52 weeks. The first 20 weeks of the combined leave shall be with benefits but without salary. After the 20-week period, the leave shall be without salary or Board contributions to benefits.

20.1.4 The teacher shall notify the Board in writing of the commencement date of maternity leave at least six weeks prior to the beginning of the leave. Where possible, the teacher should notify the Board three months prior to the commencement date. The teacher shall provide the Board with a statement from a physician indicating the anticipated date of delivery.

20.1.5 Maternity leave shall commence at the discretion of the teacher at any time prior to and including the delivery date.

20.1.6 The teacher may terminate the maternity leave with a four week, prior written notice at any time during the maternity leave period.

20.1.7 Upon completion of maternity leave, the teacher shall be assigned to the same school to which she was assigned at the commencement of the leave on the same basis as continuing members of that staff.

20.1.8 A teacher who is eligible for maternity leave under 20.1.2 is entitled to access parental leave immediately following the maternity leave as per clause 21.

20.2 Maternity Entitlements

"Maternity Supplement" means an amount paid by the Board to a teacher on a continuing contract who is not working for reasons of pregnancy or maternity. During the two week employment insurance waiting period, the maternity supplement will equal 100 percent of the teacher's regular salary. For subsequent weeks, the supplement is equal to the difference between the teacher's regular earnings and the employment insurance benefits to which the teacher is entitled.

20.2.1 Teachers with temporary, probationary or interim contracts, during the period of medical disability surrounding the delivery date, shall be entitled to access any sick leave entitlements earned. Once their sick leave entitlement is exhausted, no further sick leave or salary shall be paid.

20.2.2 Teachers with continuing contracts, who are medically disabled due to pregnancy, shall be entitled to access sick leave benefits as per clause 18.2 until the birth of the child.

(a) The Board shall pay the maternity supplement to the birth mother in the following circumstances:

(i) for a period of six weeks commencing with the birth of the child, without medical documentation

(ii) for a period beyond six weeks, if the teacher provides supporting medical documentation to the Board. This extension is subject to the restrictions of clause 18.2.

(iii) in lieu of sick leave benefits:

- when a teacher takes an unpaid, voluntary leave prior to being medically unable to work, and
- is in receipt of employment insurance benefits, and
- chooses to continue her employment insurance benefits during the period of medical disability prior to the birth.

If applicable, the supplement shall be continued beyond six weeks after the date of birth as per clause 20.2.2(a)(ii).

(b) The maternity supplement will not be paid for any days that teachers are not scheduled to work eg Christmas break, spring break and the summer.

20.2.3 No further sick leave or maternity supplement shall be paid following the 90 day elimination period for extended disability benefits.

21. Parental/Adoption Leave

21.1 Leave in accordance with clauses 21.2 to 21.6 shall be granted to a teacher who is currently on contract and who has provided one year of consecutive service under an interim, temporary, probationary and/or continuing contract in the year preceding the starting date of the leave. "Natural Breaks" as defined in clause 1(d) and periods of time up to a total of six weeks when the teacher is not under contract, shall not be considered an interruption of consecutive service in the application of clause 21.1.

21.2 Parental leave shall be available as follows:

(a) for a maximum of 37 weeks without salary or benefits,

(b) for teachers who have not accessed maternity leave, parental leave shall commence at the discretion of the teacher at any time within 52 weeks from the date of birth provided that the teacher supplies the Board with documentation of the birth,

(c) Teachers who have accessed maternity leave, shall be entitled to commence parental leave immediately following maternity leave.

21.3 Parental leave for the purpose of adoption shall be available as follows:

(a) for a maximum of 37 weeks without salary,

(b) with benefits for the first 20 weeks for one parent only,

(c) commence at the discretion of the teacher at any time within 52 weeks from the date of birth or the date the child is placed provided that the teacher supplies the Board with proof of the impending adoption.

21.4 The teacher:

(a) shall, except in extenuating circumstances, provide six weeks written notice to commence the leave,

(b) may terminate the leave with four weeks written notice at any time during the 37 week leave period.

21.5 Following parental leave, the teacher shall, upon request, be granted leave in accordance with clause 18.1.1 and 18.1.2 (d) to:

(a) a natural break in the school year,

- (b) the end of the school year in which the leave commenced,
- (c) a natural break in the next school year,

(d) the end of that school year, or

(e) any other time approved by the Board.

21.6 Upon completion of the leave granted under clause 21.2 and 21.3, the teacher shall be assigned to the school to which the teacher was assigned at the commencement of the leave, on the same basis as continuing members of that staff.

21.7 If both parents are employed by the Board, the Board may grant parental leave to both parents simultaneously.

22. Leave for Arrival of a Child

22.1 A teacher shall be granted a maximum of three operational days of leave with salary and benefits on the occasion of the birth of his/her child.

These days must be taken within two weeks of:

- (a) the date of birth; or
- (b) the day the child is released from the hospital; or
- (c) the day the mother is released from the hospital.

The teacher may choose to take the days consecutively or separately.

22.2 A teacher shall be granted a maximum of three operational days of leave with salary and benefits on the occasion of the arrival of an adopted child.

These days must be taken within two weeks of:

- (a) the date of placement; or
- (b) the day the child is released from the hospital.

The teacher may choose to take the days consecutively or separately.

23. Professional Improvement

23.1 Teachers may apply and be considered for support under both clauses 23.2 and 23.6.

23.2 Leave for professional improvement may be granted for a school year or portion thereof, upon application, to a teacher:

(a) with three or more years of service with this Board, with one experience increment but without salary or benefits,

(b) with four or more years of service with this Board, with one experience increment, benefits and an annual allowance, calculated at the time the leave commences, that is equal to step 1 on the grid for category of teacher education.

(c) with five or more years of service with this Board, with one experience increment, benefits and an annual allowance, calculated at the time the leave commences, which is the greater of 2/3 of total salary or the minimum total salary for category of teacher education,

The distribution of such leaves will not exceed 1/2 of one percent of the teaching staff in respect to any one of the categories.

23.3 The annual allowance or prorated portion thereof will be paid in equal monthly instalments over the period of the leave commencing on the last day of the first full calendar month of the leave or at the teacher's request, another mutually agreed upon arrangement between the teacher and the Board.

23.4 Not more than one experience increment can be credited while on leave or extended leave.

23.5 Teachers granted leave under clauses 23.2(b) or 23.2(c) shall:

(a) if the teachers are not in receipt of an allowance under clauses 8, 9 and 10 of this agreement, be considered during the period of leave as continuing members of the schools to which they were assigned when the leave was granted, or

(b) if the teachers are in receipt of an allowance under clauses 8, 9 and 10 for designations which continue from year to year, be granted such leaves from designation but not location.

23.6 Tuition support for professional improvement may be granted, upon application, to a teacher on continuing contract and with 2 or more years of service with this Board.

23.7 The Board will grant leaves, after reviewing the foregoing applications for leave and tuition support, to a maximum of 3/4 of one per cent of the annual grid and allowance cost, calculated as of the preceding November 30. The board will determine both the number and the persons to be granted leave and tuition support after considering the seniority of each applicant and the interests of the school system. No more than 30 per cent of the annual amount approved by the Board shall be used for tuition support.

23.8 The selection criteria and procedures established by the Board, in consultation with the Local, will be published and distributed to teachers two months prior to the deadline for receipt of applications for leaves for professional improvement and tuition support. Subsequent selection shall be made in accordance with such criteria. From time to time the Board may, in consultation with the Local, adjust the two-month requirement for publication and distribution.

23.9 In addition to professional improvement leave and tuition support provided under foregoing clauses, schools and decision units may finance the costs of short term professional improvement activities for individual teachers. These costs may include travel and living allowances and the cost of substitute teachers so that individual teachers can participate in conferences, seminars and visits to other educational jurisdictions without loss of salary.

23.10 Each school may be granted a maximum of two professional development days per school year for such activities as local professional development, inservice program planning and budgeting.

24. Liaison Committee

The liaison committee is intended to serve as a means of communication outside the context of collective bargaining. The committee shall consist of:

1) Board representatives as determined by the Superintendent of Schools and

2) the Table Officers of ATA Local 37 and other Association representatives as determined by the Local.

25. Grievance Procedure

25.1 Reference to days in this clause shall be exclusive of Saturdays, Sundays, summer break and other school holidays.

25.2 Any difference between any teacher covered by this agreement and the Board, or a dispute between the Association and the Board concerning the interpretation, operation or alleged violation of this agreement shall be dealt with as follows:

Step One—Such difference, hereinafter called a "grievance", shall be submitted in writing within 30 days of the discovery of the difference to the superintendent of schools and to the executive assistant of the Local. Such grievance shall set out the nature of the grievance, the clauses of this agreement which allegedly have been violated and the remedy being sought.

Step Two—The superintendent of schools will review the grievance and within 15 days shall render a decision in writing to the grievor and the office of the executive assistant of the Local and the coordinator of Teacher Welfare of the Association..

Optional Grievance Mediation—The parties may mutually agree to a non-binding process to attempt to resolve the grievance. To facilitate the mediation process the parties agree to extend the timeline for moving to Step Three in the grievance process so that a period of 10 days is provided after the conclusion of the meditation process to recommence formal grievance proceedings. A mediation process is attached to this agreement as a letter of intent.

Step Three—In the event that the decision of the superintendent of schools fails to satisfy the grievance, the executive assistant of the Local shall within 10 days thereafter give written notice to the Board secretary requesting consideration of the grievance by the interpretations committee.

Step Four—The interpretations committee shall be composed of two representatives of the Board and two representatives of the Association. A quorum of this committee shall consist of all members.

It shall be the duty of this committee to meet, hear the grievor and/or representative(s) and endeavor to resolve all grievances concerning the interpretation, application, operation or alleged violation of this agreement.

The interpretations committee shall be required to give its decision within 21 days following the receipt of such notice and shall dispose of each grievance before proceeding to another, except where, by unanimous consent of the interpretations committee, the hearing of such grievance is adjourned for the purpose of obtaining further information.

If the committee reaches a unanimous decision as to the disposition of any grievance, that decision shall be final and binding.

Step Five—If the committee does not reach a unanimous decision, the Association shall, within 20 days of the committee's decision, be entitled to serve the Board written notice requesting the establishment of an arbitration board.

Each party shall appoint one member as its representative on the arbitration board within seven days of such notice and the two members so appointed shall endeavor to select an independent chairperson. If the two members fail to select a chairperson within five days after the date on which the last of the two members is appointed, they shall request the Director of Mediation Services to select a chairperson.

The arbitration board shall determine its own procedures but shall give full opportunity to all parties to present evidence and to be heard.

The arbitration board shall not change, modify or alter any of the terms of this agreement. All grievances or differences submitted shall present an arbitrable issue under this agreement and shall not depend on or involve an issue or contention by either party that is contrary to any provisions of this agreement or that involves the determination of subject matter not covered by, or arising during the term of this agreement.

The arbitration board shall give its decision within a reasonable time after the appointment of the chairperson. A decision of a majority of the members of the arbitration board shall govern but, if there is no majority decision, the

decision of the chairperson shall be the decision of the arbitration board. In either situation, the decision is binding on both parties.

Each party to arbitration shall bear the expense of its respective nominee and the two parties shall bear equally the expenses of the chairperson.

25.3 By mutual agreement between the Board and the Association, any of the dates contained in this clause may be extended but, in the absence of mutual agreement, failure by the grieving party to comply with any of the time restrictions will mean that the grievance is lost.

26. Subrogation

26.1 (a) Cost of Absence means the total remuneration paid by the Board during a period when the teacher was absent from work.

(b) Interest means interest calculated in accordance with the provisions of the Alberta Judgement Interest Act, RSA 2000, c.J-1, and amendments and regulations thereto.

(c) Judgement or Settlement means an order of a court of competent jurisdiction or an agreement whereby the teacher agrees to accept any sum of money representing past or future loss of remuneration, either by lump sum, periodic payment(s), or through the purchase of an annuity, or any of them.

(d) Remuneration means the salary, allowances, benefit premiums and other monies paid to or in respect of the teacher by the Board

(e) Teacher means a teacher in respect of whom the Board has incurred a cost of absence and includes the teacher's personal representative, trustee, guardian or the estate of the deceased teacher.

26.2 In the event that the Board incurs a cost of absence as a result of an act or omission of a third party, the Board is subrogated to any right of recovery of the teacher from the third party in the amount of the cost of absence and without restricting the generality of the foregoing, the following provisions apply:

(a) the teacher shall advise the Board in advance of the teacher's intention to initiate any claim in which an act or omission of a third party has resulted in the Board incurring a cost of absence;

(b) the teacher shall upon request by the Board include the cost of absence, as calculated by the Board, in the teacher's claim;

(c) the Board shall have the right (but not the obligation) to maintain an action in the name of the teacher and engage a solicitor (including the teacher's solicitor) to recover the cost of absence;

(d) the teacher agrees to cooperate with the Board and to provide, at the Board's expense, all loss of income records, transcripts, loss of income reports and information with respect to the calculation or allocation of damages and attend examinations for discovery or assist as a witness where required;

(e) the teacher will not settle his/her claim without the prior written consent of the Board as to the amount of the cost of absence to be recovered by the Board;

(f) upon resolution of the amount of the cost of absence payable to the Board, the Board may, upon default of payment by the teacher following demand by the Board offset the agreed upon amount of the cost of absence payable to the teacher by the Board; (g) the teacher shall not release any third party from the cost of absence without the consent of the Board; and

(h) the Board's consent to settlement shall not be unreasonably withheld.

26.3 When as a result of judgement or settlement with the consent of the Board, the teacher recovers a sum equal to all of the cost of absence, the teacher shall, as of the date of settlement or judgment, pay the full cost of absence recovered to the Board plus interest, less a proportionate share of legal fees payable thereon by the teacher to his/her solicitor with respect to such recovery.

26.4 When as a result of a judgement or settlement with the consent of the Board, the teacher recovers a sum equal to a portion of the cost of absence, the teacher shall as of the date of settlement or judgment, pay to the Board, the amount of the cost of absence recovered plus interest, less a proportionate share of legal fees payable thereon by the teacher to his/her solicitor with respect to such recovery.

26.5 The teacher will upon request by the Board execute such documents and agreements as may be required or deemed desirable by the Board to give effect to the provisions of this clause 26.

26.6 In exercising any of its rights under clause 26., the Board shall have due regard for the interests of the teacher.

LETTER OF INTENT: Extracurricular Activities

With respect to clause 13 of the collective agreement:

The Board recognizes and values the many volunteer hours that teachers commit to school-related activities beyond their instructional assignments. The parties recognize that teachers make this commitment voluntarily in a spirit of professionalism, and with the desire to contribute to students, families and the life of the school community.

With respect to the 370 minutes per week of assignable time provided for in clause 13.2, this will be assigned in a reasonable manner for the kinds of activities specified. Teachers will not be assigned duties on weekends, holidays or unusual hours.

LETTER OF INTENT: Framework for Involvement in Site-Based Decision Making

The parties hereby agree that the joint committee (maximum of 10 members) consisting of an equal number of representatives from the Board and the Local, shall continue to investigate and report on the current status and application of the Framework for Involvement in Site-Bases Decision Making (May 2000). The committee shall continue to collaboratively monitor and evaluate the effectiveness of the Framework for Involvement in Site-Based Decision Making on an ongoing basis and provide recommendations regarding the future use of the Framework for Involvement in Site-Based Decision Making within the district. The committee shall issue a written report to the Superintendent of Schools and the President of the ATA Local by March 1 of each year.

LETTER OF INTENT: Guidelines for Classroom Safety

The parties hereby agree that within 90 days of the signing of this collective agreement, a Joint EPSB/ATA Occupational Health and Safety Committee, consisting of an equal number of representatives from the Board and the Local, shall be formed to examine workplace health and safety issues related to teaching staff.

The Terms of Reference shall be developed by the committee and will include, but not be limited to:

Completion of the development and implementation of guidelines for "Classroom Safety of Students in Science and Career Technology Studies (CTS)";

Monitoring and evaluation of the effectiveness of the guidelines for "Classroom Safety of Students in Science and Career Technology Studies (CTS)", once implementation is completed;

Hazard assessments related to teaching;

Safety procedures when required to work alone, as outlined in the provincial legislation;

Related staff education materials.

Minutes of the committee meetings and written recommendations shall be provided to the Local and the Board on an ongoing basis.

LETTER OF INTENT: Argyll Centre

The parties hereby agree that within 30 days from the date of the exchange of opening positions, the joint committee (maximum of 10 members) consisting of an equal number of representatives from the Board and the Local, shall

examine teaching assignments at Argyll Centre. The committee shall provide a report to the Superintendent of Schools and the President of the Local by no later than January 31, 2008.

LETTER OF INTENT: Non-Binding Grievance Mediation Process

(a) After receipt of the decision of the Superintendent of Schools, under Step Two, within 10 working days either party may request that a Mediator be appointed to meet with the parties, investigate and define the issues in dispute and facilitate a resolution.

(b) The mediator shall be appointed by mutual agreement between the parties.

(c) The purpose of the Mediator's involvement in the grievance process is to assist the parties in reaching a resolution of the dispute, and anything said, proposed, generated or prepared for the purpose of trying to achieve a settlement is to be considered privileged. During the proceedings, the parties shall disclose all materials and information relevant to the issue(s) in dispute.

(d) The expenses of the Mediator shall be equally borne by both parties.

(e) The grievance may be resolved by mutual agreement between the parties. Within 10 working days of first meeting the parties, having considered the issue(s) in dispute and the terms of the collective agreement, the Mediator shall issue a report including non-binding recommendations.

LETTER OF INTENT: Sick Leave

The parties agree to form a committee to analyze the utilization and implementation of sick leave. The committee shall include representatives from the School Board, The Alberta Teachers' Association, and ATA Local 37. Additionally, the committee shall request representation from the Alberta School Employee Benefit Plan. The intent of the committee will be to report on ways to promote health in the workplace, review strategies to manage recurring illnesses, and identify

opportunities to reduce the exposure of the district to benefit premium surcharges while still maintaining benefit entitlements of teachers. Such committee shall meet at least once before December 31, 2007.

LETTER OF INTENT: Benefits

The District does not have any intention of moving teacher benefit plans from the Alberta School Employee Benefit Plan. If, prior to August 31, 2012, the administration believes that there would be an advantage to examining alternate benefit carriers and engaging their services if they can provide equivalent or better benefits at a cost equal to or less than ASEBP, the administration will provide notice to the representatives of the Association. Prior to making a recommendation regarding alternate benefits carriers, the administration will enter into discussions with the Local and will examine solutions and options for addressing this matter. A decision to change carriers will require ratification by both the Board and the Local.

LETTER OF UNDERSTANDING

The Parties understand and intend that the changes in salaries, rates of pay, allowances and substitute pay referenced in this Memorandum of Settlement effective on September 1 of each 2007, 2008, 2009, 2010 and 2011 are exactly the same as those intended by the Province of Alberta and the ATA stated in their November 15, 2007 Memorandum of Agreement. In the event of conflict between the interpretation of the November 15, 2007 memorandum and this Memorandum of Settlement regarding the changes to salaries, rates of pay, allowances and substitute pay, the terms of the November 15, 2007 memorandum shall govern and take precedence.

ADDENDUM TO THE 2007-2012 COLLECTIVE AGREEMENT BETWEEN EDMONTON SCHOOL DISTRICT NO 7 AND THE ALBERTA TEACHERS' ASSOCIATION

Rates Effective September 1, 2008

5.1 Salary Scale

Years of teaching experience Years of University Education

	Four	Five	Six
0	50,999	53,920	57,223
1	53,907	56,828	60,132
2	56,815	59,738	63,040
3	59,723	62,646	65,948
4	62,631	65,554	68,856

5	65,540	68,462	71,764
6	68,448	71,370	74,672
7/8	71,356	74,278	77,580
9	74,264	77,186	80,488
10	77,172	80,094	83,397
11	80,693	83,616	86,918

7.1 Substitutes – The rate of pay for substitute teachers shall be, inclusive of general holiday and vacation pay, as follows:

Full \$188 Day

Half \$105 Day

9.1 Principals – The formula to be used in computing the principal's allowance for each school will be as follows:

Minimum \$21,533 + \$15.35 (P-300)

Maximum \$33,616

9.3 Curriculum Coordinators and Department Heads - \$6,740

Assistant Curriculum Coordinator and Assistant Department Heads - \$4,519

9.4 Teacher Consultants - \$6,470

10.1 Directors receive a salary equal to the maximum principal allowance provided for in clause 9.1 plus the maximum salary for six years provided for in the salary grid in clause 5.1 - \$120,534.

10.2 Supervisors receive a salary equal to 90 per cent of a director's salary - \$108,480.

10.3 Twelve Month Consultants receive a salary equal to 90 per cent of a supervisor's salary - \$97,632.

10.4 Program Coordinators receive a salary equal to a maximum assistant principal's allowance plus the maximum salary provided for on salary grid in clause 5.1 - \$103,726.

11.1 Teachers Assigned to Multiple Locations – A teacher whose assignment requires travel between buildings separated by at least one kilometer shall be paid an annual allowance as follows - \$1,243.

If a teacher is regularly scheduled to teach in more than 2 buildings in a single day, there shall be an additional allowance as follows - \$1,243.

11.2 Teachers at Night School and Summer School – A teacher employed on an hourly basis to provide instruction in credit courses at the Board's night school and summer school shall be paid at a rate inclusive of general holiday and vacation pay of - \$56.98.

11.3 Work Experience, Registered Apprenticeship Program Coordinator - \$2,719.

15.2 In a school where there is no assistant principal, a teacher shall be named to carry out administrative duties during any absence of the principal.

The principal may, for the school year, designate up to two teachers who are not in receipt of any other allowances to serve for the school year as surrogate principal during the principal's absence. There shall be a surrogate principal's annual allowance of \$1,292.

ADDENDUM 2

COLLECTIVE AGREEMENT BETWEEN BOARD OF TRUSTEES EDMONTON SCHOOL DISTRICT NO 7 AND THE ALBERTA TEACHERS' ASSOCIATION SEPTEMBER 1, 2007 TO AUGUST 31, 2012

5. Salary Scale

5.1 Effective September 1, 2009 teachers will be paid in accordance with the following salary grid.

Years of teaching experience Years of University Education

	Four	Five	Six
0	54,054	57,150	60,651
1	57,136	60,232	63,734
2	60,218	63,316	66,816
3	63,300	66,398	69,898
4	66,383	69,481	72,980
5	69,466	72,563	76,063
6	72,548	75,645	79,145
7	75,630	78,727	82,227
8/9	78,712	81,809	85,309
10	81,795	84,892	88,392

5.4.3 e) Effective September 1, 2009, a teacher shall, upon initial placement on the salary grid, be granted only one additional increment during any one school year, based on experience as defined in clause 5.4.1.

7.1 Substitutes – The rate of pay for substitute teachers shall be, inclusive of general holiday and vacation pay, as follows:

Full \$199 Day

Half \$115 Day

9.1 Principals – The formula to be used in computing the principal's allowance for each school will be as follows:

Minimum \$22,823 + \$16.27 (P-300)

Maximum \$35,630

9.3 Curriculum Coordinators and Department Heads - \$6,858

Assistant Curriculum Coordinator and Assistant Department Heads - \$4,790

9.4 Teacher Consultants - \$6,858

10. Salaries for Designated Positions (12-months)

10.1 Directors receive a salary equal to the maximum principal allowance provided for in clause 9.1 plus the maximum salary for six years provided for in the salary grid in clause 5.1 - \$127,754.

10.2 Supervisors receive a salary equal to 90 per cent of a director's salary - \$114,978.

10.3 Twelve Month Consultants receive a salary equal to 90 per cent of a supervisor's salary - \$103,480.

10.4 Program Coordinators receive a salary equal to a maximum assistant principal's allowance plus the maximum salary provided for on salary grid in clause 5.1 - \$109,939.

11. Other Allowances

11.1 Teachers Assigned to Multiple Locations – A teacher whose assignment requires travel between buildings separated by at least one kilometer shall be paid an annual allowance as follows - \$1,317.

If a teacher is regularly scheduled to teach in more than two buildings in a single day, there shall be an additional allowance as follows - \$1,317.

11.2 Teachers at Night School and Summer School – A teacher employed on an hourly basis to provide instruction in credit courses at the Board's night school and summer school shall be paid at a rate inclusive of general holiday and vacation pay of - \$60.39.

11.3 Work Experience and/or Registered Apprenticeship Program Coordinator - \$2,882.

LETTERS OF INTENT

Amend

"Framework for Involvement in Site-Based Decision Making"

The parties hereby agree that the joint committee (maximum of 10 members) consisting of an equal number of representatives from the Board and the Local, shall continue. The committee working on the Framework for Involvement in Site-Based Decision Making [May 2000] shall expand its focus to investigate and report on developing a culture that engages, values and respects staff, fosters their well being and enables them to be involved in decision making processes resulting in staff satisfaction. This committee will explore strategies and best practices that can be used by schools to achieve these ends. The committee shall issue a written report to the Superintendent of Schools and the President of the ATA Local at least once within each school year.

New

"Early Intervention Pilot Program"

The parties hereby agree to participate in the Alberta School Employee Benefit Plan's Early Intervention Pilot Program (EIPP). The purpose of the program is to facilitate the teacher's early return to work from illness or injury on a part-time basis. Participation in the EIPP is strictly voluntary on the part of individual teachers.

If a teacher agrees in writing to participate in the EIPP, sick leave days in clauses 18.2.2, 18.2.3 and 18.2.4 shall be interpreted using modified formula. Days where the teacher is at work for a full day do not count as a sick day. Days where the teacher is at work for a portion of the day will count as a partial day of sick leave so that the portion of the day worked and the portion of the day reported as sick leave together equal one full day. Days where the teacher does not work, including days not scheduled for work, count as full sick days.

If a teacher agrees to participate in the EIPP and provides written notice to withdraw, the teacher will be eligible to access their remaining sick leave entitlement on a full-time basis.

Either party may choose to withdraw from the program using the procedure outlined in the "Early Intervention Plan Pilot Program Agreement".

ADDENDUM 3

COLLECTIVE AGREEMENT BETWEEN BOARD OF TRUSTEES EDMONTON SCHOOL DISTRICT NO 7 AND THE ALBERTA TEACHERS' ASSOCIATION SEPTEMBER 1, 2007 TO AUGUST 31, 2012

5. Salary Scale

5.1 Effective September 1, 2010 teachers will be paid in accordance with the following salary grid.

Years of teaching

Years of University Education

experience

	Four	Five	Six
0	55,632	58,819	62,422
1	58,804	61,991	65,595
2	61,976	65,165	68,767
3	65,148	68,337	71,939
4	68,321	71,510	75,111
5	71,494	74,682	78,284
6	74,666	77,854	81,456
7	77,838	81,026	84,628
8	81,010	84,198	87,800
9/10	84,183	87,371	90,973
11	88,024	91,213	94,814

7. Substitute Teachers

7.1 The rate of pay for substitute teachers shall be, inclusive of general holiday and vacation pay, as follows:

Full \$205 Day

Half \$123 Day

9.1 Principals – The formula to be used in computing the principal's allowance for each school will be as follows:

Minimum \$23,489 + \$16.75 (P-300)

Maximum \$36,670

9.3 Curriculum Coordinators and Department Heads - \$7,058

Assistant Curriculum Coordinator and Assistant Department Heads - \$4,930

9.4 Teacher Consultants - \$7,058

10. Salaries for Designated Positions (12-months)

10.1 Directors receive a salary equal to the maximum principal allowance provided for in clause 9.1 plus the maximum salary for six years provided for in the salary grid in clause 5.1 - \$131,484.

10.2 Supervisors receive a salary equal to 90 per cent of a director's salary - \$118,336.

10.3 Twelve Month Consultants receive a salary equal to 90 per cent of a supervisor's salary - \$106,502.

10.4 Program Coordinators receive a salary equal to a maximum assistant principal's allowance plus the maximum salary provided for on salary grid in clause 5.1 - \$113,149.

11. Other Allowances

11.1 Teachers Assigned to Multiple Locations – A teacher whose assignment requires travel between buildings separated by at least one kilometer shall be paid an annual allowance as follows - \$1,355.

If a teacher is regularly scheduled to teach in more than two buildings in a single day, there shall be an additional allowance as follows - \$1,355.

11.2 Teachers at Night School and Summer School – A teacher employed on an hourly basis to provide instruction in credit courses at the Board's night school and summer school shall be paid at a rate inclusive of general holiday and vacation pay of - \$62.15.

11.3 Work Experience and/or Registered Apprenticeship Program Coordinator - \$2,966.

15.2 Administrative Absences – There shall be a surrogate principal's annual allowance of \$1,409.